Author's response to reviews

Title: The e-Health Implementation Toolkit: Qualitative evaluation across four European countries

Authors:

Anne MacFarlane (anne.macfarlane@nuigalway.ie)
Pauline Clerkin (pauline.clerkin@nuigalway.ie)
Elizabeth Murray (Elizabeth.murray@pcps.ucl.ac.uk)
David F Heaney (d.heaney@abdn.ac.uk)
Mary Wakeling (mary@bongosoft.co.uk)
Ulla-Maija Pesola (umpesola@informatik.umu.se)
Eva Lindh Waterworth (eva@informatik.umu.se)
Frank Larsen (frank.larsen@telemed.no)
Minna Makiniemi (Minna.Makiniemi@ppshp.fi)
Ikka Winblad (ilkka.winblad@oulu.fi)

Version: 4 Date: 28 October 2011

Author's response to reviews:

October 28th 2011

RE: The e-Health Implementation Toolkit: Qualitative evaluation across four European countries Journal: Implementation Science

MS ID: 9913010853482741

Dear Editor,

Thank you for your e-mail of October 20th and I am now submitting a revised version of our manuscript based on your guidance in that e-mail.

The following major revisions have been made:

• Author list: The list of authors in the manuscript is written exactly as they are in the submission system, both in style and order.

• The authors' names are now in the preferred style for your journal i.e. 'First name Initial Last name'.

• A functioning link to the website on page 7 is now included.

• All references are now cited in the text using consecutive numbers in square brackets. Citations in tables follow on from the last citation in the main text (not where the table is to be placed).
• In the authors’ contributions, we have indicated that all authors read and approved the final manuscript.

The following minor revisions have been made:

• The text 'Title:' has been removed from the title.

• Each author’s country has been included in the affiliation details.

• The reference list has been edited so that it contains up to 30 authors’ names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.

• The titles 'Figure legends' and 'Tables' have been added in above the respective sections.

• The visible vertical lines have been removed from all tables.

• Figure 1 has been cropped as closely as possible to minimise white space around the image and has been converted to .tiff format.

• The manuscript has been checked for any typographical errors and final corrections or revisions have been made.

Please let me know if there is anything else I need to do to complete work on this manuscript.

Yours sincerely,

Anne MacFarlane, Ph.D

Lecturer in Primary Care, Discipline of General Practice, NUI Galway

Galway, Ireland